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Project Definition:

The ILC Controls effort is on multiple fronts: 

1) Providing a controls system for the SCRF R&D program at Fermilab. This includes NML, Horizontal Test Stand(s), Vertical Test Stand(s) at 1B1. It also includes the controls system for 3.9GHz cavity testing as many of the components being developed at used at A0 as many of these will be moved/deployed at NML. 
2) Global ILC effort. We will help write the RDR including cost estimates. We will help write the EDR and participate in controls R&D necessary for its completion. We will help participate in instrumentation R&D working on HOM BPM measurements and ATCA digitizers.

3) Cavity Data Management System. 
Project Manager’s Summary:
ILC: 

NML: 
Network fiber is pulled to the control room and experimental floor.  Switch installation will commence after the fiber has been terminated.  UPS circuits and power distribution in the control room complete.

A0:
The laser, gun, and 9-cell were successfully run with the new 9MHz clock/timing system.  Beam to the dump was established.  Commissioning of the new master oscillator and SIMCON llrf boards continues.
Updated DOOCs to run under SLF5.  Changed the EPICS and DOOCs build systems to build position independent code.  Implemented a simple BPM logger and phase scan script.   

Esecon board software development is underway.

Set up the timing systems for 1.3 GHz and 3.9 GHz systems to use the same configuration.
HTS:
Coupler conditioning on 1.3 GHz cavity was completed successfully.  Digitizer problems were resolved.   A new edm screen for HTS was developed and is being used.
Infrastructure:
A WIKI has been set up for documenting the work on this project.  It is available at:

http://www-bd.fnal.gov/issues/wiki/ILCControls.   Meeting minutes, agendas and other information can be accessed here.

Research was done into the GIT source control management system as well as others to replace the CVS system that we are using.  We are looking for something that will aid in collaboration between laboratories.   Began work on SLF5 migration.

ATCA:
The Motorola F101 switch blade stopped functioning and was sent back to Motorola for repair.
Open Clovis training was held September 25-27 on site.    

Resources Used in September 2007 (FTEs):

	
	AD
	CD
	TD
	PPD
	Total Effort

	ILCTA
	1.69
	3.55
	0.00
	0.35
	5.59

	ILC
	0.34
	0.49
	0.00
	0.00
	0.83

	DM
	0.00
	0.00
	0.00
	0.00
	0.00


The effort listed here is time worked and does not include vacation, sick leave, holidays, etc.

Purchase requisitions/procard during September, 2007:  

Vacuum system PLC hardware
Hytec blade with one scaler IP and one Analog to Digital IP for nml Xray counters
Misc. components for RF interlocks
Milestones:

Meetings held, Reports Given:

Status meetings were held  Sep 10th, and 17th.
NML Controls Priority and Schedule meeting  September 24, 2007 at 1:30 pm
New Documents posted to ILC DOCDB: 

473-v1 Monthly Report of the ILC Controls Project for August, 2007
441-v1  Horizontal Test Stand Control System   

